Jane Figueiredo, PhD
Samuel Lunenfeld Research Institute
Mount Sinai Hospital
60 Murray Street Box 18
Toronto, ON M5T 3L9 CANADA
jane@mshri.on.ca

RE: 572327718122179

Dear Dr. Edmunds:

Please find our final revised manuscript "Polymorphisms eMyc-N11S and p27-V109G and breast cancer risk and prognosis" enclosed for your further consideration. We have made the formatting revisions as requested.

Specific formatting edits:

1. We have removed the capitalization of the manuscript title (page 1)
2. We have removed 'running head', 'keywords', 'article type', 'disclosures by authors..' from page 1.
3. We have removed qualifications (i.e., Ph.D.) from author details (page 1).
4. We have made the following changes: page 2: Abstract - Introduction → Background, page 3: INTRODUCTION → Background, page 4: MATERIALS AND METHODS → Methods, page 8: RESULTS → Results, page 10: DISCUSSION → Discussion
5. We have included a statement of ethical approval (page 5).

Thank-you for considering our work. Please contact me if you have any additional concerns.

Sincerely,

Jane Figueiredo